

Economics and Business Quarterly Reviews

Njaala, P. D. (2023). Does Privatisation Improve Financial and Operating Performance of a Firm? The Case of Tanzania Breweries Ltd. *Economics and Business Quarterly Reviews*, 6(4), 74-94.

ISSN 2775-9237

DOI: 10.31014/aior.1992.06.04.540

The online version of this article can be found at: https://www.asianinstituteofresearch.org/

Published by: The Asian Institute of Research

The *Economics and Business Quarterly Reviews* is an Open Access publication. It may be read, copied, and distributed free of charge according to the conditions of the Creative Commons Attribution 4.0 International license.

The Asian Institute of Research *Economics and Business Quarterly Reviews* is a peer-reviewed International Journal. The journal covers scholarly articles in the fields of Economics and Business, which includes, but is not limited to, Business Economics (Micro and Macro), Finance, Management, Marketing, Business Law, Entrepreneurship, Behavioral and Health Economics, Government Taxation and Regulations, Financial Markets, International Economics, Investment, and Economic Development. As the journal is Open Access, it ensures high visibility and the increase of citations for all research articles published. The *Economics and Business Quarterly Reviews* aims to facilitate scholarly work on recent theoretical and practical aspects of Economics and Business.





The Asian Institute of Research Economics and Business Quarterly Reviews Vol.6, No.4, 2023: 74-94 ISSN 2775-9237 Copyright © The Author(s). All Rights Reserved DOI: 10.31014/aior.1992.06.04.540

Does Privatisation Improve Financial and Operating Performance of a Firm? The Case of Tanzania Breweries Ltd

Papias Daudi Njaala¹

¹Lecturer, Faculty of Accounting and Finance, Institute of Accountancy Arusha, Njiro Hill, P.O. Box 2798, Arusha, Tanzania. Phone: +255 784 436 286. E-mail: papiasnjaala@gmail.com and pnjaala@yahoo.co.uk

Abstract

Extant theoretical and empirical literature show that profitability, efficiency, capital expenditure, output and dividends improve (increase) after privatisation. However, previous studies showed a mixture of results on employment levels after privatisation: some showing increases and some showing decreases in employment levels after privatisation. Previous studies also showed that leverage levels decrease after privatisation. In this study the author reviewed some literature regarding privatisation initiatives carried out in various countries and their repercussions on financial and operating performance of firms being privatised. He then used the case of Tanzania Breweries Ltd and compared its pre- and post-privatisation performance levels to evaluate whether there was any post-privatisation improvement. The t-test Paired Two Sample for Means available in Data Analysis Tool of Ms Excel was used to calculate pre- and post-privatisation means (averages) and mean differences for selected performance metrics and also for hypotheses testing. The results show that profitability (in terms of return on sales, return on assets and return on equity), operating efficiency (in terms of sales efficiency and net income efficiency), capital expenditure (in terms of capital expenditure to sales), output (in terms of hectolitres produced and real sales), dividend payout (in terms of dividend to sales ratio and dividend per share) and earnings in terms of earnings per share improved (increased) significantly after privatisation. Capital expenditure in terms of capital expenditure to assets increased but there was no strong evidence to support that the increase was attributable to privatisation or it was just by chance. Financial leverage (in terms of debt to assets and debt to equity) and number of employees improved (decreased) significantly after privatisation.

Keywords: Privatisation, Financial Performance, Operating Performance, State-Owned-Enterprises, SOEs, Performance Metrics

1. Introduction

"If you want to know the end, look at the beginning." (African Proverb)

1.1. Background of the Study

Since coming into power of Margaret Thatcher in United Kingdom in 1979 the wave of privatisation swept market, socialist and mixed economies. Normally state-owned enterprises (SOEs) face inefficiencies originating

from politicians' interference leading into excess employment, operating inefficiencies (Chen *et* al, 2021), low capital investment, low output, high operating costs and huge losses or unreasonably low profit levels, high gearing and low dividends (Čučković, *et al*, 2011). Due to inefficiencies (Čučković *et* al, 2011) and lack of profitability resulted from politicians' interference on SOEs, governments in both developed and developing countries embarked in privatisation programmes to heal unhealthy SOEs from their maladies (Čučković, *et* al, 2011; Redda, 2007; Bachiller, 2017 and Kisenge, 2013). The wave of privatisation started to sweep developed countries to the larger extent than developing countries but in 1990s we evidenced an important shift in both the industries being privatised and in the number of countries which participated in privatisation. For example, in 1990s and earlier 2000s we witnessed more privatisations in many sectors including highly regulated industries such as banking and telecommunications and electronic utilities (Čučković, *et* al, 2011) in both developed and developing countries, and privatisation programmes are clearly spreading throughout the developing economies. The percentage of privatisations in developing countries as a proportion of total global privatisations increased (Megginson *et al*, 2001).

Privatisation in this paper is taken as broadly defined as the deliberate sale by a government of SOEs or assets to private economic agents (Megginson *et al*, 2001). It is also defined as the sale of a state-owned firm to private sector (Cuervo *et al*, 2000). However, the sale of a state-owned firm can be wholly or partial in the sense that the government may sell all of its stake or a portion thereof.

Extant empirical studies show that privatisation of SOEs results into significant improvement in corporate performance (Chen, *et* al, 2021, Čučković, *et* al, 2011; Dewenter *et al*, 2001 and D'Souza, *et al*, 2005). However, there is a mixture of results on employment levels resulting from privatisation. Bortolotti, *et al* (2001) pointed out that privatisation improves significantly profitability, output, operating efficiency and capital investment spending whereas employment and leverage decline significantly as a result of privatisation.

Extant privatisation studies carried out in Tanzania focused on privatization process and asset valuation (Waigama, 2008), the role of privatisation in providing solid waste collection services (Kaare, 2002), performance of privatised (Mongula, 2002) and the impact of privatisation of a bank (Kisenge, 2013).

A case of Tanzania Breweries Ltd (TBL) was taken to assess performance improvement emanating from privatisation. Ratios were calculated and analysed from two sets of TBL's financial statements and their implications are explored. The first set covered the periods 1985-1992 and the second one covered 1994-2001. The year 1993 was excluded from the analysis because it was the year in which TBL was privatised. In this study the author examined financial and operating performance of TBL to illustrate whether privatisation of SOEs improves corporate performance. Corporate performance in the context of this paper focuses on corporate profitability, operating efficiency, capital investment, output improvement, gearing, payment of dividends to shareholders and earnings available to shareholders.

Profitability analysis presented and analysed includes the use of profitability ratios which are return to sales = net profit after tax/total sales revenue; return on assets = net profit after tax/total sales revenue; and return on equity = net profit after tax/total equity.

Operating performance is presented and analysed in terms of ratios namely sales efficiency = total sales revenue/total number of employees, and net income efficiency = net profit after tax/total number of employees. Capital investment spending ratios analysed and presented include capital expenditure to sales= capital expenditure/sales; and capital expenditure to total assets = capital expenditure/total assets.

Output improvement analysis includes analysis of sales in physical term = hectolitres of beer produced, and sales (in real terms) = nominal sales x (consumer price index_{Reference year}/consumer price index_{Base year}).

Analysis of gearing ratios includes analysis of debt to assets = total debt/total assets and debt to equity = total debt/total equity whereas dividend analysis includes dividend per share = total dividend paid/number of ordinary shares issued and dividend to sales = dividend/total sales revenue.

Earnings available to shareholders is analysed in terms of earnings per share = profit after tax/number of issued ordinary shares.

1.2. Reasons for Privatisation of SOEs

Čučković, *et al* (2011), Megginson *et al* (2001), Shirley (1992), Bachiller (2017), Bachiller (2012), Bartolotti (2001), Dewenter and Malatesta (2001), Estrin and Pelletier (2018), Megginson and Netter (2001), Balza et al (2013), Arocena and Oliveros (2012), Mongula (2002) and Chen *et al* (2021) are some of researchers who provided some arguments in support of privatisation programmes and policies being implemented by governments mostly since 1980's. Some of their reasons for embracing privatisation programmes are discussed here below:

1.2.1. Presence of insignificant Market Failure

Megginson *et al* (2001) pointed out that privatisation has positive impact and hence appealing in areas where market failure is noticeably insignificant, that is market is competitive or can become readily competitive. In the same vein they argue that privatisation is less appropriate for public goods and natural monopolies since competitive considerations are weaker in these respects.

1.2.2. Contracting Ability by Owners in Private Firms impacts the Efficiency

The government ownership of firms results in problems in defining the goals of the firm. They pointed out that while the shareholder-wealth-maximising model of corporate organisation is becoming increasingly dominant, in part because of the advantage of having a well-defined corporate goal; governments have many objectives other than profit or shareholder-wealth-maximisation (Čučković, *et al*, 2011; Megginson *et al*, 2001; and Shirley, 1992). Further, government objectives can change from one regime to the next hence affecting the SOEs. Megginson *et al* (2001) also pointed out that in a public enterprise, owners (citizens) are less able to influence managers to act in their interests which is not the case in a privately-owned enterprise.

1.2.3. The Cost-Benefits and Easy with which Government can intervene in Firm's Operations

Čučković, *et al* (2011) and Megginson *et al* (2001) pointed out that government can intervene in the operations of any firm being public or private; however, it is more costly for the government to intervene in a private firm than in a SOE. Thus, they argued, that to the extent that government intervention has greater costs than benefits, private ownership is preferred to public ownership.

1.2.4. Less-prosperous Firms Reliance on Government for Funding leading to "Soft" Budget Constraints

The government is unlikely to allow a large SOE to become bankrupt and as such the discipline enforced on private firms by capital markets and the threat of financial distress is less important for SOEs (Shirley, 1992; Megginson and Netter, 2001; Kouser *et al*, 2011).

1.2.5. Proceeds from Privatisation can impact Efficiency through its Effect on Government Fiscal Conditions

Privatisation in some cases has been used as a fiscal instrument to cover (reduce) fiscal deficit through selling SOEs (Megginson and Netter, 2001; Čučković, *et al*, 2011; Shirley, 1992). They pointed out that proceeds from privatisation of SOEs have helped reduce fiscal deficit in many countries. They also pointed out that in some instances proceeds from privatisation of SOEs are saved by governments instead of being spent.

1.2.6. Privatisation improves operational Efficiency

As a results of privatisation operational efficiency in terms of output per employee, sales per employee, net income per employee, and sales in real terms improves (Čučković, *et al*, 2011; Kouser *et al*, 2011; Arocena and Oliveros, 2012; Bachiller, 2012, Bachiller, 2017; Mongula, 2002; Estrin and Pelletier, 2018 and Balza *et al*, 2013).

1.2.7. Improvement of productivity

Most studies on productivity showed post-privatisation productivity improvement to differing extents (Chen *et al*, 2021; Mongula, 2002 and Kouser *et al* (2011).

1.2.8. Privatisation can help develop Factor, Product and Security Markets and Institutions

The privatisation has a role to play in development of factor and product markets and security markets (Čučković, *et al*, 2011; Megginson and Netter, 2001). Thus, to the extent that privatisation promotes competition, it can have important role to play in enhancement of efficiency; but at the same time the strength of markets prior to privatisation also has an impact on the success (or otherwise) on post-privatisation performance, Megginson and Netter (2001).

1.2.9. Profitability improvement

One of the main motives for privatisation is that empirical evidence showed that post-privatisation profitability improved significantly (Kouser *et al*, 2011; Čučković, *et al*, 2011; Bachiller, 2017; Bartolotti *et al*, 2001; Dewenter and Malatesta, 2001; Estrin and Pelletier, 2018; Megginson and Netter, 2001 and Balza *et al*, 2013).

1.3. The Aim of the Study

The aim of the study is to find the empirical evidence on whether privatisation results into improvement on financial and operating performance of a firm in Tanzania. Extant empirical literature shows that no similar study on post-privatisation financial and operating performance at micro/ firm level has been carried out in Tanzania. The rest of this paper is organised as follows: Section 2 provides the literature review, section 3 provides data and methodology used, section 4 presents and discusses research results (findings) and section 5 provides the concluding remarks and areas for further researches.

2. Literature Review

"However far a stream flows, it doesn't forget its origin." (Nigerian, African Proverb) "The child you sired hasn't sired you." (Somalia, African Proverb)

In this section the researcher visits the work of previous researchers on privatisation and its effects to SOEs. Theoretical and empirical studies regarding the multi-national and multi-industry privatisation; case study and industry-specific privatisation; and empirical research in privatisation are visited.

2.1. Multi-national and multi-industry Research on Performance of privatised Firms

The effect of privatisation on financial and operating efficiency for multi-national and multi-industry has been researched by various writers including (Chen *et al*, 2021; Earle and Shpak, 2019; Al Hinai, 2016); Megginson and Netter (2001); and Dewenter and Malatesta (2001).

Chen, *et al* (2021) carried out a study on how ownership affects productivity of SOEs in China and found that on average privately owned firms are 53% more productive than SOEs, but the benefits of privatization take several years to fully materialize. They also found that the said productivity gap is smaller among larger firms and

economically more liberal times (later period of privatisation) and places (South and Coast China) than in economically less liberal times (beginning of privatisation) and places (North and Inland) and it is larger in consumer-facing and high-tech industries. In other words, they found that privatisation improved productivity level to greater extent in small and medium enterprises than in larger ones and in more economically liberal times and places than elsewhere. In the same vein they also found that productivity improvement was more noticed in sectors dealing with final goods and high-tech than "heavy" industries.

The existence of SOEs is associated with excess employment, excess budget spending for unnecessary subsidies, lack of profits or unreasonably low profits (Earle and Shpak, 2019; Al Hinai, 2016), lack of incentives for technological and skills improvements, low output and productivity levels, low wages (Earle and Shpak, 2019). However, regarding employment, Earle and Shpak (2019) found that SOEs are generally associated with excess employment for low-skilled workers. Al Hinai (2016) undertook a study on comparison between pre- and post-privatisation financial and operating performance of 61 Egyptian companies over a period of 16 years and a comparison between those 61 privatized SOEs and private companies over 16 years. The method of privatisation for the companies was initial public offerings (IPO). Findings of Al Hinai (2016) showed significant positive post-privatisation change in profitability, and operating efficiency and a negative impact on the leverage and employment level. The comparison of post-privatisation results to those of private firms revealed that there was significant increase in EBIT and ROE with no significance for ROS and ROA. For total debt to total equity ratio there was no significant improvement while employment level change was significant.

Balza *et al* (2013) explored the relationship between private sector participation, institutional reform, and performance of firms in electricity sector in 18 Latin American countries in the span of four decades. Their findings suggested that privatisation results into improvements in quality and efficiency rather than accessibility to electricity service. They also found that the quality and stability of regulatory institutions was strongly associated with accessibility to service regardless of the level of private sector participation.

Arocena *et al* (2012) undertook a study on post-privatisation effect on efficiency in Spain: comparing privatised firms and their counterparts before privatisation. Using data envelopment analysis (DEA) and regression analysis, they found that there were no significant differences in efficiency between the SOEs and privatised firms, the efficiency of private counterparts improved significantly and the efficiency of competitors during the same post-privatisation showed no significant improvement.

Arcas and Bachiller (2010) studied the role of organisational changes and contextual factors in affecting the operating performance of private European companies. Their findings show that smaller and non-regulated firms and private firms through public offer perform better than bigger, regulated, and privatised by private sale method. They also found that privatised Eastern European companies were less profitable than counterparts from other European countries. Their findings also suggest that organisational changes are more difficult to introduce in privatisation through private sale and in bigger and regulated companies. Arcas and Bachiller (2010) also suggest that less developed institutional context limit post-privatisation performance improvement.

Bachiller, (2012) undertook research on privatisation of 38 European firms and found that performance of utility firms improved significantly after privatisation. However, there was no sufficient evidence to conclude whether post-privatisation of firms in other sectors improved or not.

Dewenter and Malatesta (2001) found and documented from their study of privatisation effect of 63 companies privatised during the period from 1981 to 1993. They documented increases on profitability measured as return on sales. However, their study documented that profitability as measured by earnings before interest and taxes to sales declined insignificantly as a result of privatisation. With their sample they also documented that privately-owned companies are more profitable and efficient than SOEs.

Bartolotti, D'Souza, Fantini and Megginson (2001) carried out a study in 25 countries on 31 telecommunication firms privatised during the period from October 1981 to November 1998 and found that post-privatisation profitability, output, operating efficiency and capital investment spending increase significantly, whereas

employment and leverage decline significantly. Their results show that over two-thirds of all firms in the sample experienced a post-privatisation increase in operating income to sales of 74.1%, ROS by 70.4%, ROA by 66.7%. These increases are significant at 1%, 5% and 10% levels of significant. However, they found that 59.3% of the firms experienced insignificant increase in ROE.

2.2. Case Study and Industry-specific Privatisation Research

The effect of privatisation on financial and operating efficiency in the form of case study and industry-specific privatisation researches were carried out by various writers including Čučković, *et al* (2011), Branston (2000); Buckland and Frasser (2002); and Portelli and Narula (2004). These studies involved a single firm or a small number of firms in a single industry. The researchers here normally used econometrics to study the effect of privatisation of SOEs.

Čučković, et al (2011) investigated the impact of marketization and privatisation of the telecommunication (Croatian Telecom (HT)) and gas and oil industries (gas company INA) in Croatia in order to determine whether privatisation improves business efficiency, employment levels, investment, service quality, diversity and prices or not. They found that HT maximised net proceeds badly needed for the budget of the country and offered a good investment opportunity for domestic institutional investors, mainly the state and private pension funds, as well as small shareholders through the offer of substantial price discounts to former and new employees. This resulted into improved liquidity at Zagreb Stock Exchange as it contributed significantly to the general volume of transactions. The study also found that sales revenue initially remained stagnant amid increased competition, regulation, restructuring of the market and diversification of services (from 2004 to 2006). Later on, revenues from fixed telephony declined by 37%, revenues from mobile telephony increase by 17% and revenues from internet services increased by 184% (from 2006 to 2009). Subsequently (after 2009), HT's market share in internet services and mobile services started to decrease slightly due to increased competition resulting into a slight decline in sales revenue in these business segments. Čučković, et al (2011) also pointed out that HT's improved post-privatisation performance was associated with falling levels of employment by 31% (from 8,862 to 6,116 employees). However, HT's labour productivity (revenue per employee) improved by 56% as a result of HT's privatisation. The HT's productivity gains were mainly achieved through substantial reduction of labour and increase in sales revenue after privatisation. This substantial and dramatic drop in the number of employees was unpopular, and was often disputed and resisted by the trade unions. The new HT's management dealt with this dissatisfaction with a generous severance pay plan, and by providing training and in-house assistance for job search for laid-off workers. In contrast, HT Mobile (a separate entity within the T-HT Group), increased the number of employees by 3% in 2003-2009, but this did not alter the general downward trend in the number of employees in the company as a whole. The study (of Čučković, et al (2011)) also found that HT's earnings ratio, profit margin improved significantly initially but later they deteriorated due to world economic crisis, stringent competition and pressures related to regulations and fiscal policies. They also found that post-privatisation investment rate in infrastructure and other long-term capital assets increased.

Croatian privatisation of gas sub-sector commenced on 2002. Čučković, *et al* (2011) also found that after being privatized to Hungarian oil company MOL, the Croatian gas company INA's revenues almost doubled. However, the company could achieve higher revenue increases in case the tight price increase restrictions imposed by the Croatian Energy Regulatory agency (CENA) and central government were not there. Regarding post-privatisation level, INA maintained employment level at around 16,000 (from 2002 to 2008) until 2010 when the company announced restructuring and employment reduction by 9%. This suggests that there was a small influence on employment levels after privatisation. However, the maintenance of employment levels was due to a clause in the Privatisation Agreement with MOL which required INA not to lay workers off within a five-year period after privatisation. Čučković, *et al* (2011) also found that post-privatisation labour productivity improved even without layoffs. In this regard revenue per employee doubled in 2004 - 2008 from €118,999 to €225,950, declining only in 2009 to €170,265 per employee as a result of global economic crisis. Čučković, *et al* (2011) points out that INA together with its Italian partners made significant post-privatisation investments in the modernization of refineries in Rijeka and Sisak, especially in 2008 – 2009. This significant investment in long-term assets led to post-privatisation improvement of sales to long-term assets could be higher than that

level if MOL was initially given majority shares in INA and if the government did not impose price restrictions on oil and gas. The government price restrictions led to low profit levels necessitating the investments in longterm assets to be made from borrowed funds rather than re-invested funds from retained earnings leading to a steep surge in the debt/equity ratio. Earnings and earnings to net sales also fell in 2008 and recovered slightly in 2009 and more significantly in 2010 (for earnings) and significantly (for earnings to net sales) in 2009, Čučković, *et al* (2011). The initial INA's disappointing post-privatisation results were caused by factors such as government price restrictions, poor corporate governance (with management and supervisory board members appointed by government on political basis).

2.3. Other previous Empirical Findings on Privatisation

Other previous empirical findings on privatisation were documented by various researchers. They include (Earle and Shpak, 2019; Estrin and Pelletier (2018); Redda, 2007; etc.). Extant documented empirical evidence shows that privatisation alone does not automatically improve operating and financial performance of a privatised firm. Various authors have studied the conditions driving the performance of a firm after being privatised. They include Radić *et al* (2021), Earle and Shpak (2019), Estrin and Pelletier (2018), Čučković, *et al* (2011), Redda (2007) and Megginson, *et al* (2006).

According to Estrin and Pelletier (2018) and Čučković, *et al* (2011), Radić *et al* (2021), for privatisation to work (i.e., produce positive results) it should be associated with robust regulatory infrastructure and appropriate process of privatization. These include well-designed and sequenced reforms; the implementation of complementary policies; the creation of regulatory capacity; attention to poverty and social impacts; and strong public communication.

For post-privatisation financial and operating performance of a SOE to improve, firms should be subjected to competition (Earle and Shpak, 2019; and Estrin and Pelletier, 2018), good candidates be chosen (Radić *et* al, 2021), good privatisation process be used (Radić *et* al, 2021), business environment should be improved (Earle and Shpak, 2019) and quality and stable regulatory institutions should be put in place (Radić *et* al, 2021). They pointed out that privatisation that leads to private monopoly results into worse performance than that before privatisation (Earle and Shpak, 2019; and Estrin and Pelletier, 2018; Radić *et* al, 2021).

Čučković *et al* (2011) points out that privatization of monopolistic SOEs assisted the introduction competition by allowing the entry of new firms (through rules and regulation) to the telecommunication market that had previously been dominated by state monopoly. As a result of privatization, quantity and quality of services improved, transparency of consumers' rights for timely and correct information about services was enhanced, inland calls on fixed telephone lines prices (which were underpriced through subsidies) increased and foreign calls, internet access and other services prices (which were overpriced) decreased after privatisation. Postprivatisation prices reflect underlying operating costs, investment costs, competition level and benefits to consumers. Ultimate result of privatization led to both improved profitability to telecommunication service providers and improved customer satisfaction in Croatia.

According to Redda (2007) for a privatised firm's operating and financial performance to improve the following factors should exists: (a) which firms are privatized; there can be a positive (or negative) selection effect; (b) whether privatization is total or partial; evidence suggests that the former is more beneficial; (c) the regulatory framework, which in turn depends on the institutional and political environment; (d) the characteristics of the new owners; foreign ownership has been associated with superior post-privatization business performance, especially relative to "insider" ownership (privatization to managers and workers) and (e) effective competition. Redda (2007) pointed out that competition had been found to be critical in bringing about improvements in company performance because it is associated with lower costs, lower prices, and higher operating efficiency.

Bachiller (2017) applied a meta-analysis of 60 empirical studies of privatised firms. The aim was to evaluate whether different post-privatisation results stem from the method of privatisation and the level of development of a country in which privatised firm existed. She found that firms privatised using public offerings perform

better after privatisation than those privatised using other methods such as voucher privatisation and private sale. The study also refuted the common-place assumption that privatisation in developing countries does not improve financial performance.

Waigama (2008) studied implementation of privatisation process and valuation methodology of privatised SOEs in Tanzania in order to assist the Presidential Parastatal Sector Reform Commission (PSRC) in decision-making. The study traced five interrelated aspects of privatisation, namely strategy formulation, valuation method, valuation error, assessment of buyer of SOEs and post-privatisation developments. The study attempted to find out whether or not the five aspects proceeded in the way to attain the privatisation stated objectives. The study found that PSRC's privatisation strategy did not promote higher competition, higher prices and higher government revenue; valuation methodology applied by valuation consultants did not improve certainty in determination of reserve price, and valuation estimates were not good proxies of sale prices; wider ownership participation by people was not achieved and follow up on ownership changes and post-privatisation methodology to produce intended results the market system and its institutions should be well developed and function well. These were lacking in privatisation process and valuation methodology in Tanzania (Waigama, 2008).

Kaare (2002) studied the impact of privatisation of solid waste collection services in Dar es Salaam, Tanzania and found that the privatisation had minimal impact in refuse collection services to low-income households. Mongula (2002), carried out a qualitative study of privatised firms in Tanzania and found that post-privatisation performance for some firms improved while for some other firms did not improve. The firms with improved post-privatisation performance were Tanzania Breweries Ltd., Tanzania Distilleries, Darbrew Ltd., Tanzania Cables, Tanzania Cigarettes Co., Tanzania Portland Cement Co., Tanga and Mbeya Cement factories, National Bank of Commerce (1997) Ltd., ABB Tanelec (electrical equipment manufacturers), Morogoro Canvas, Sabuni (Foma) Detergents, Kibo Paper Industry, DAHACO and Cam and Metal Box and Mtibwa, Kilombero and TPC sugar factories. Post-privatisation improvement involved tax payments, increase in production, acquisition of new technology and managerial skills, and levels of salaries. The firms whose post-privatisation performance did not improve included Tanzania Shoe Company and Pollysacks Ltd, Ubungo Spinning (has been closed since March 2002), Burns and Blane, National Bicycle Company (NABICO), Tanganyika Dyeing and Weaving Mill (Sungura Textile), Tanzania Pharmaceuticals Co., HANDICO and Blankets and Textiles Manufacturers. Mongula (2002) provided some reasons for improvement (or otherwise) of the firms: privatisation process, history of the privatised firms before privatisation, post-privatisation governance and management and decisionmaking, financial constraints of post-privatisation owners, organised resistance by employees (related to postprivatisation salary levels and working environment/ standards). The study by Mongula (2002) involved multisector firms and was narrative in nature without the use of any data analysis tool. Thus, even for firms that portrayed post-privatisation improvements the study did not provide any evidence whether the improvement was significant or not.

Kisenge (2013) conducted a qualitative study on the impact of privatising the National Bank of Commerce (NBC) of Tanzania and found that the bank's post-privatisation improved in terms of the volume of loans provided, job creation, increase number of branches which brought services near citizen, new technologies which facilitate the use of bank services without being in the bank premises (for example the use ATM machine, internet banking), online payment services (such as *LUKU*, telephone credits and payroll processing service).

The hitherto studies conducted in Tanzania did not relate to financial and operating performance metrics and methodology used in the current study. This provided the gap that the author attempted to plug. Therefore, this study focuses to plug the gap by evaluating the effect of privatisation on financial and operating performance of Tanzania Breweries Ltd.

Post-privatised TBL is not a monopolist firm, it is regulated by Dar es Salaam Stock Exchange (DSE) and Registrar of Companies and other government regulatory institutions such as Fair Competition Commission (FCC), and good business environment therefore it is predicted that its post-privatisation results to reflect

improved financial and operating results. The empirical TBL's pre- and post-privatisation financial and operating results are presented and discussed in section 4.0 of this paper.

3. Data and Methodology

"If you do not have patience, you cannot make beer." (Namibia, Ovambo tribe African Proverb)

3.1. Introduction

This section provides research design, data and data analysis and hypothesis tested.

3.2. Research Design

This study has used descriptive research design that is useful for a study involving both quantitative and qualitative approaches. The design provides descriptive data such as minimum values, maximum values, means, variances, standard deviation, correlation, t-statistics, p-values and critical values. In this study the descriptive research design has been used to describe the effect of privatisation on financial and operating performance of TBL, and test a variety of hypotheses on whether privatisation results into improvement of financial and operating performance. The study is confirmatory because it attempts to test hypotheses that privatisation improves financial and operating performance of SOEs as documented in previous studies. It is carried out to support or refute the hypotheses. Specifically, it tests the hypotheses that profitability, operating efficiency, capital expenditure, output, dividend payout and earnings per share of TBL improved (increased) and financial leverage and employment improved (decreased) after privatisation.

3.3. Data and Data Analysis

Data collected and used in the study were sales revenue, profit after tax, total assets, total equity, number of employees, capital expenditure, output (in hectolitres of beer produced), total debt, dividend paid, number of shares issued and earnings per share.

Data for the study were mainly collected from TBL Annual Reports from the year 1985 to 1992 (preprivatisation period), that is year -8 to -1 and 1994 to 2001 (post-privatisation period), that is year +1 to +8. Some data were also obtained from the internet, for example from World Bank reports. Year of privatisation 1993, defined as Year 0 is excluded from calculating the pre- and post-privatisation means and mean differences.

Table 3.1 shows proxies used and analysed for profitability, operating efficiency, capital expenditure, output, financial leverage, employment level, dividend payout and earnings. Proxies for profitability are return on sales (ROS) ratio, return on assets (ROA) ratio and return on equity (ROE) ratio. Proxies for operating efficiency are sales efficiency (SALEFF) and net income efficiency (NIEFF). Capital expenditure to sales (CESA) and capital expenditure to assets (CETA) are used as proxies for capital expenditure. Proxies for output are units produced (in hectolitres) (PROD) and real sales (RESA). Financial leverage is proxied by debt to assets (TDTA) ratio and debt to equity (TDTE) ratio. The proxy for employment level is the number of employees (EMPL) whereas proxies for dividend payout are dividend to sales (DIVSAL) and dividend per share (DPS). Earnings are proxied by earning per share (EPS).

In order to test the predicted changes indicated in tables 3.1 and 3.2, the study used the t-test Paired Two Sample for Means available in Data Analysis Tool of Ms Excel. This was used to test whether mean differences between pre- and post-privatisation were zero.

Regarding real sales calculations, nominal sales (in TZS M) were adjusted for inflation by using the following formula, as also applied by Kenton (2022):

Real Sales = Nominal Sales x $\underline{CPI}_{Reference year}$

CPI_{Base year}

Where:

$$\label{eq:Real-Sales} \begin{split} \text{Real Sales=Inflation-adjusted sales} \\ \text{CPI}_{\text{Reference year}} = A \text{ consumer price index for a reference year} \\ \text{CPI}_{\text{Base year}} = A \text{ consumer price index for a base year} \end{split}$$

After calculating means and differences of means for each variable, p-values are used to test whether the changes in financial and operating performance indicators are statistically significant. In other words, the test is carried out to determine whether the difference averages between pre- and post-privatisation samples are zero.

The t-test Paired Two Sample for Means data analysis was carried out using Microsoft Excel Data Analysis Tool. The tool is used because it is useful when: (i) testing two measurements on the same company or companies, (ii) two sample sizes are equal, (iii) sample sizes are small (i.e., less than 30 items), (iv) sample observations are not completely independent but their dependent in pairs, and (v) simple random sampling is used, Sancheti and Kapoor (2007). The technique is applicable, for example when a company or a government institutes a policy/ intervention to improve performance of a company or companies and an analysis is required to assess whether the policy/ intervention worked effectively. All the mentioned conditions for the use of the method/ technique were present for the case in hand, that is TBL.

Regarding hypothesis testing, null hypothesis was formulated as, H_0 : There is no difference between pre- and post-privatisation performance indicator (difference of means) and alternative hypothesis was formulated as, H_A : There is an increase or decrease in post-privatisation performance indicator (difference of means). The hypotheses to be tested are shown in table 3.2. The aim is to reject the null hypothesis in case the post-privatisation improvement is statistically significant, that is if the p-value is less that significance level used. The significance level used in this case is 0.05.

3.4. Hypotheses Tested

This section presents all hypotheses tested in this study. Table 3.2 shows these hypotheses. Generally, the hypotheses test whether TBL's privatisation improved financial and operating performance. The hypotheses are presented here below:

3.4.1. Hypotheses on Profitability

(a) Privatisation had no effect on return on sales (ROS)

 H_{O} : $ROS_{Pre} - ROS_{Post} = 0$

 H_A : $ROS_{Pre} - ROS_{Post} < 0$

A null hypothesis (H_O) stated that the difference (change) between pre-privatisation ROS and post-privatisation ROS is zero. In other words, the null hypothesis stated that privatisation did not affect ROS. The alternative hypothesis (H_A) stated that privatisation improves (increases) ROS.

- (b) Privatisation had no effect on return on assets (ROA)
- H_{O} : $ROA_{Pre} ROA_{Post} = 0$

 H_A : $ROA_{Pre} - ROA_{Post} < 0$

A null hypothesis (H_O) stated that the difference between pre-privatisation ROA and post-privatisation ROA is zero. In other words, the null hypothesis stated that privatisation did not affect ROA. The alternative hypothesis (H_A) stated that privatisation improves (increases) ROA.

(c) Privatisation had no effect on return on equity (ROE)

 H_{O} : $ROE_{Pre} - ROE_{Post} = 0$

 H_A : $ROE_{Pre} - ROE_{Post} < 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation ROE and post-privatisation ROE is zero. In other words, the null hypothesis stated that privatisation did not affect ROE. The alternative hypothesis (H_A) stated that privatisation results into improvement of ROE (increasing ROE).

3.4.2. Hypotheses on Operating Efficiency

(a) Privatisation had no effect on sales efficiency (SALEFF)

 H_{O} : SALEFF_{Pre} - SALEFF_{Post} = 0

 $H_A: \qquad SALEFF_{Pre} - SALEFF_{Post} < 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation SALEFF and post-privatisation SALEFF is zero. In other words, null hypothesis stated that privatisation did not affect SALEFF. The alternative hypothesis (H_A) suggested that privatisation improves (increases) SALEFF.

(b) Privatisation had no effect on net income efficiency (NIEFF)

 H_{O} : NIEFF_{Pre} - NIEFF_{Post} = 0

H_A: NIEFF_{Pre} – NIEFF_{Post} < 0

A null hypothesis (H_0) stated that the difference between pre-privatisation NIEFF and post-privatisation NIEFF is zero. In other words, null hypothesis stated that privatisation did not affect NIEFF. The alternative hypothesis (H_A) suggested that privatisation improves (increases) NIEFF.

3.4.3. Hypotheses on Capital Expenditure

(a) Privatisation had no effect on capital expenditure to sales ratio (CESA)

 H_{O} : CESA_{Pre} – CESA_{Post} = 0

 H_A : CESA_{Pre} – CESA_{Post} < 0

A null hypothesis (H_0) stated that the difference between pre-privatisation CESA and post-privatisation CESA is zero. This is to say, the null hypothesis stated that privatisation did not affect CESA. The alternative hypothesis (H_A) suggested that privatisation improves (increases) CESA.

(b) Privatisation had no effect on capital expenditure to assets (CETA) ratio

 H_{O} : CETA_{Pre} – CETA_{Post} = 0

 $H_A: \qquad CETA_{Pre} - CETA_{Post} < 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation CETA and post-privatisation CETA is zero. In other words, null hypothesis proposes that privatisation did not affect CETA. The alternative hypothesis (H_A) stated that privatisation improves (increases) CETA.

3.4.4. Hypotheses on Output

(a) Privatisation had no effect on units produced in hectolitres (PROD)

 H_{O} : $PROD_{Pre} - PROD_{Post} = 0$

 $H_{A}: \qquad PROD_{Pre} - PROD_{Post} < 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation PROD and post-privatisation PROD is zero. This is to say, null hypothesis stated that privatisation did not affect PROD. The alternative hypothesis (H_A) suggested that privatisation improves (increases) PROD.

(b) Privatisation had no effect on real sales (RESA)

 H_{O} : RESA_{Pre} - RESA_{Post} = 0

H_A: $RESA_{Pre} - RESA_{Post} < 0$

Null hypothesis (H_0) proposes that the difference between pre-privatisation RESA and post-privatisation RESA is zero. In other words, null hypothesis proposes that privatisation did not affect RESA. Alternative hypothesis (H_A) suggested that privatisation improves (increases) sales in real terms (RESA).

3.4.5. Hypotheses on Financial Leverage

(a) Privatisation had no effect on debt to assets (TDTA) ratio

H_O: $TDTA_{Pre} - TDTA_{Post} = 0$

 $H_A: \qquad TDTA_{Pre} - TDTA_{Post} > 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation TDTA and post-privatisation TDTA is zero. In other words, the null hypothesis stated that privatisation did not affect TDTA. The alternative hypothesis (H_A) predicted that privatisation improves (decreases) TDTA.

(b) Privatisation had no effect on debt-to-equity ratio (TDTE)

H_O: $TDTE_{Pre} - TDTE_{Post} = 0$

 H_A : $TDTE_{Pre} - TDTE_{Post} > 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation TDTE and post-privatisation TDTE is zero. This is to say, the null hypothesis stated that privatisation did not affect TDTE. The alternative hypothesis (H_A) stated that privatisation improves (decreases) TDTE.

3.4.6. Hypothesis on Employment Level

(a) Privatisation had no effect on the number of employees (EMPL)

 H_{O} : $EMPL_{Pre} - EMPL_{Post} = 0$

H_A: $EMPL_{Pre} - EMPL_{Post} > 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation EMPL and post-privatisation EMPL is zero. This null hypothesis said that privatisation did not affect the number of employees. The alternative hypothesis (H_A) said that privatisation resulted into decrease in the number of employees.

3.4.7. Hypotheses on Dividend Payout

(a) Privatisation had no effect on the dividend to sales ratio (DIVSAL)

- H_{O} : DIVSAL_{Pre} DIVSAL_{Post} = 0
- $H_{A}: \qquad DIVSAL_{Pre} DIVSAL_{Post} < 0$

A null hypothesis (H_0) indicated that the difference between pre-privatisation DIVSAL and post-privatisation DIVSAL was zero. In other words, the null hypothesis said that privatisation did not affect the DIVSAL. The alternative hypothesis (H_A) said that privatisation resulted into improvement (increase) of DIVSAL.

(b) Privatisation had no effect on the dividend per share (DPS)

H_O: $DPS_{Pre} - DPS_{Post} = 0$

H_A: $DPS_{Pre} - DPS_{Post} < 0$

A null hypothesis (H_0) said that the difference between pre-privatisation DPS and post-privatisation DPS was zero. The null hypothesis indicated that privatisation did not affect the DPS. The alternative hypothesis (H_A) stated that privatisation resulted into improvement (increase) of DPS.

3.4.8. Hypothesis on Earnings

- (a) Privatisation had no effect on the earnings per share (EPS)
- H_O: $EPS_{Pre} EPS_{Post} = 0$
- $H_A: EPS_{Pre} EPS_{Post} < 0$

A null hypothesis (H_0) stated that the difference between pre-privatisation EPS and post-privatisation EPS was zero. The null hypothesis said that privatisation did not affect the EPS. The alternative hypothesis (H_A) said that privatisation resulted into improvement (increase) of EPS.

4. Research Results: TBL's Pre- versus Post-Privatisation Financial and Operating Performance

"The new moon cannot come until the other has gone" (Bahunde or Hunde, Democratic Republic of Congo Proverb)

4.1. Introduction

The results of the study are presented and analysed in this section. As already mentioned in section 3.0 of this paper, *t*-test Paired Two Sample for Means was carried out to determine whether privatisation of TBL improved its financial and operating performance. Tables 3.1 and 3.2 indicate performance metrics used to assess the effect of privatisation. The metrics fall under the following categories: profitability, efficiency, capital expenditure, output, financial leverage, employment, dividend payout and earnings.

	Performance Metrics	Definitions	Predicted Change after Privatisation
1.		Profitability	
(i)	Return on Sales (ROS)	Profit after tax/Total sales revenue	Increase, i.e. ROS _{Pre} < ROS _{Post}
(ii)	Return on Assets (ROA)	Profit after tax/Total assets	Increase, i.e. $ROA_{Pre} < ROA_{Post}$
(iii)	Return on Equity (ROE)	Profit after tax/Total equity	Increase, i.e. ROE _{Pre} < ROE _{Post}
2.		Operating Efficiency	
(i)	Sales Efficiency (SALEFF)	Total sales revenue/Number of employees	Increase, i.e. $SALEFF_{Pre} < SALEFF_{Post}$
(ii)	Net Income Efficiency (NIEFF)	Profit after tax/Number of employees	Increase, i.e. NIEFF _{Pre} < NIEFF _{Post}
3.		Capital Expenditure	
(i)	Capital Expenditure to Sales (CESA)	Capital expenditure /Total sales revenue	Increase, i.e. CESA _{Pre} $<$ CESA _{Post}
(ii)	Capital Expenditure to Assets (CETA)	Capital expenditure/Total assets	Increase, i.e. $CETA_{Pre} < CETA_{Post}$
4.		Output	
(i)	Units produced in hectolitres (PROD)	Total hectolitres of beer produced	Increase, i.e. PROD _{Pre} < PROD _{Post}
(ii)	Real Sales (RESA)	Nominal Sales x <u>CPI_{Reference year}</u> CPI _{Rase year}	Increase, i.e. RESA _{Pre} < RESA _{Post}
5.		Financial Leverage	
(i)	Debt to Assets (TDTA)	Total debt/Total assets	Decrease, i.e. TDTA _{Pre} > TDTA _{Post}
(ii)	Debt to Equity (TDTE)	Total debt/ Total equity	Decrease, i.e., TDTE _{Pre} >TDTE _{Post}
6.		Employment	
(i)	Number of Employees (EMPL)	Number of eemployees	Decrease, i.e. EMPL _{Pre} > EMPL _{Post}
7.		Dividend Payout	
(i)	Dividends to Sales (DIVSAL)	Dividends/Total sales revenue	Increase, i.e., DIVSAL _{Pre} < DIVSAL _{Post}
(ii)	Dividend Per Share (DPS)	Dividends/ No. of issued ordinary shares	Increase, i.e., DPS _{Pre} < DPS _{Post}
8.		Earnings	
(i)	Earnings per share (EPS)	Profit (Loss) after tax/No. of issued ordinary shares	Increase, i.e., EPS _{Pro} < EPS _{Port}

Table 3.1: Performance Metrics, Definitions and Predicted Changes

Source: Author's formulation

Key for table 3.1 above:

ROS _{Pre}	stands for return on sales before privatisation
ROS _{Post}	stands for return on sales after privatisation
ROA _{Pre}	stands for return on assets before privatisation
ROA _{Post}	stands for return on assets after privatisation
ROE _{Pre}	stands for return on equity before privatisation
ROE _{Post}	stands for return on equity after privatisation
SALEFFPre	stands for sales efficiency before privatization
SALEFFPost	stands for sales efficiency after privatization
NIEFFPre	stands for net income efficiency before privatization
NIEFFPost	stands for net income efficiency after privatization
CESA _{Pre}	stands for capital expenditure to sales before privatisation
CESA _{Post}	stands for capital expenditure to sales after privatization
CETA _{Pre}	stands for capital expenditure to assets before privatisation
CETA _{Post}	stands for capital expenditure to assets after privatisation
PROD _{Pre}	stands for production output (in hectolitres) before privatisation
PROD _{Post}	stands for production output (in hectolitres) after privatisation
RESA _{Pre}	stands for real sales before privatisation
RESA _{Post}	stands for real sales after privatisation
TDTA _{Pre}	stands for total debt to assets before privatization
TDTA _{Post}	stands for total debt to assets after privatization
TDTE _{Pre}	stands for total debt to equity before privatization
TDTE _{Post}	stands for total debt to equity after privatization
EMPLPre	stands for number of employees before privatisation
EMPLPost	stands for number of employees after privatisation
DIVSALPre	stands for dividend to sales before privatization
DIVSALPost	stands for dividend to sales after privatization
DPS _{Pre}	stands for dividend per share before privatization
DPSPost	stands for dividend per share after privatization
EPS _{Pre}	stands for earnings per share

Table 3.2: Predicted Change and Hypotheses tested

	Performance Metrics	Predicted Change after	Difference	Hypotheses tested							
1		Privatisation									
1.		Profitability	200 200	H D							
	Return on Sales (ROS)	Increase, i.e.,	ROS _{Pre} - ROS _{Post}	H_0 : Difference = 0							
		$ROS_{Pre} < ROS_{Post}$		H_A : Difference > 0							
	Return on Assets (ROA)	Increase, i.e.,	ROA _{Pre} - ROA _{Post}	H_0 : Difference = 0							
		$ROA_{Pre} < ROA_{Post}$		H_A : Difference > 0							
	Return on Equity (ROE)	Increase, i.e.,	ROE_{Pre} - ROE_{Post}	H_0 : Difference = 0							
-		ROE _{Pre} < ROE _{Post}		H_A : Difference > 0							
2.	Operating Efficiency										
	Sales Efficiency (SALEFF)	Increase, i.e.,	SALEFF _{Pre} - SALEFF _{Post}	H_0 : Difference = 0							
		SALEFF _{Pre} < SALEFF _{Post}		H_A : Difference > 0							
	Net Income Efficiency (NIEFF)	Increase, i.e.,	NIEFF _{Pre} - NIEFF _{Post}	H_0 : Difference = 0							
2		NIEFF _{Pre} < NIEFF _{Post}		H_A : Difference > 0							
3.		Capital Expenditu	re	H D'00 0							
	Capital Expenditure to Sales (CESA)	Increase, i.e.,	CESA _{Pre} - CESA _{Post}	H_0 : Difference = 0							
		CESA _{Pre} < CESA _{Post}		H_A : Difference > 0							
	Capital Expenditure to Assets (CETA)	Increase, i.e.,	CETA _{Pre} - CETA _{Post}	H_0 : Difference = 0							
4		$CETA_{Pre} < CETA_{POst}$		Π_A . Difference > 0							
4.	Units produced in heatelitres (BBOD)		PROD PROD	$\mathbf{H} \cdot \mathbf{Diff}$ orongo = 0							
	Units produced in nectonities (FROD)	$PROD_{-} < PROD_{-}$	FRODPre - FRODPost	H ₀ . Difference ≥ 0							
	Real Sales (RESA)	Increase i.e.	RESA RESA.	H_A : Difference = 0							
	Real Sales (RESA)	RESA _D < RESA _D	RESApre - RESApost	H ₀ : Difference ≥ 0							
5	I	Financial Leverag	e	II _A . Difference > 0							
	Debt to Assets (TDTA)	Decrease i e	TDTA _P , -TDTA _P ,	H_0 : Difference = 0							
		$TDTA_{Pre} > TDTA_{Pret}$	i Dilipre i Dilipost	H ₀ : Difference < 0							
	Debt to Equity (TDTE)	Decrease i e	$TDTE_{P_{m}}$ - $TDTE_{P_{m+1}}$	H_0 : Difference = 0							
	Deer to Equily (1212)	$DTE_{Pre} > DTE_{Post}$		H_{A} : Difference < 0							
6.		Employment									
	Number of Employees (EMPL)	Decrease, i.e.	EMPL _{Pre} - EMPL _{Post}	H_0 : Difference = 0							
		$EMPL_{Pre} > EMPL_{Post}$	110 1000	H_A : Difference < 0							
7.		Dividend Payout		•							
	Dividends to Sales (DIVSAL)	Increase, i.e.,	DIVSAL _{Pre} - DIVSAL _{Post}	H_0 : Difference = 0							
		$DIVSAL_{Post} < DIVSAL_{Post}$		H_A : Difference > 0							
	Dividend per Share (DPS)	Increase, i.e.,	DPS _{Pre} - DPS _{Post}	H_0 : Difference = 0							
		$DPS_{Pre} < DPS_{Post}$		H_A : Difference > 0							
8.		Earnings									
	Earnings per share (EPS)	Increase, i.e.	EPS _{Pre} - EPS _{Post}	H_0 : Difference = 0							

	$EPS_{Pre} < EPS_{Post}$	H_A : Difference > 0

Source: Author's formulation

Profitability metrics used are ROS, ROA and ROE. Efficiency metrics used are sales efficiency and net income efficiency. The capital expenditure metrics are capital expenditure to sales and capital expenditure to assets whereas output metrics gauged are units produced (in hectolitres) and real sales. The author gauged debt to assets and debt to equity as financial leverage metrics and number of employees as employment metrics. Dividends are measured in terms of dividend to sales and dividend per share whereas earnings are measured in terms of earnings per share.

4.2. Change in Profitability after Privatisation

Tables 4.1 and 4.2 show annual performance metrics for eight years prior to privatisation and eight years after privatisation of TBL. The minimum pre-privatisation ROS stood at -223.06% whereas the minimum post-privatisation ROS was -4.31%. The maximum pre-privatisation ROS was 5.85% whereas the maximum post privatisation ROS was 28.51%. The average pre-privatisation ROS was -118.80% and the average post-privatisation ROS stood at 15.05% (refer to table 4.2). With *t*-stat of -3.995847748, t-critical value (one-tail) of 1.894578605 and p-value (one-tail) of 0.002608712 (which is less than significance level of 0.05) (refer to table 4.2), null hypothesis is rejected. This means that difference between the pre-privatisation average ROS and the post-privatisation average ROS (increase) is statistically significant. In other words, privatisation of TBL had significant positive impact on return on sales.

Performance Metrics	1985	1986	1987	1988	1989	1990	1991	1992	1994	1995	1996	1997	1998	1999	2000	2001
Profitability																
Tionability																
ROS (%)	-223.06	-238.87	-161.04	-211.19	-1.24	-78.3	5.85	-42.54	-4.31	12.88	13.49	11.97	14.6	18.62	28.5	24.65
ROA (%)	-21.27	-18.72	-17.26	-23.55	-0.24	-14.59	1.41	-7.69	-3.99	21.73	30.38	20.99	22.56	25.32	31.33	27.23
ROE (%)	-42.72	-40.14	-33.73	(43.90)	-0.4	-22.83	2.24	-11.6	-5.54	30.2	43.13	39.71	44.83	47.79	46.19	35.03
Operating Efficiency																
SALEFF (TZS M)	0.392459 0	0.3369713 5	0.5478991 6	0.604225 352	1.149076 517	1.1628622 72	1.713624 67	1.91206 896	11.19034 74	25.16926 3	59.24048 33	73.073587 39	76.63857 80	91.116173 1	82.063636 3	87.65801 527
NIEFF (TZS M)	- 0.875409 84	0.8049113	0.8823529 4	- 1.2760 563	0.014248 02	- 0.9104898 6	0.100257 06	0.81344 827	- 0.482625 48	3.242989 47	7.992145 01	8.7495400 79	11.19124 4	16.966590 7	23.393181 8	21.61068 702
Capital Expenditure																
CESA (TZS M)	166.6666 66	162.34817 8	107.87321 06	113.9860 14	61.01033 295	47.419497 13	54.50045 00	74.9323 7151	27.70589 65	21.27741 43	9.112328 26	9.4580014 03	8.760973 766	15.2125	5.3958494 8	3.689738 052
CETA (TZS M)	15.89134 93	12.726118 7	11.564178 45	12.71120 354	11.89080 331	8.8350113 01	13.17736 67	13.5474 405	25.61403 50	26.37799 71	9.035369 77	16.577101 88	13.53550 1	16.129397 7	5.9297960 8	0.795444 708
Output																
PROD (Hectolitres)	520,545	531,127	530,475	529,955	495,355	435,347	469,523	453,548	575,000	875,000	1,221,307	1,450,000	1,500,00 0	1,535,700	1,551,00	1,545,00 0
RESA (TZS M)	39.78828 00	54.364700	111.88320 00	160.9608 00	411.2862 00	385.79806 0	729.9270	739.703 0	6,495.09 03	17,072.31 12	33,873.85 65	44,609.539 80	52,677,3 350	58,584.000	56,014.340 4	62,434.1 584
Financial Leverage																
TDTA (%)	79.24	0	0	45.49	40.83	24.4	21.25	17.72	17.14	0	9.27	15.45	10.96	8.21	8.05	10.75
TDTE (%)	159.17	0	0	84.79	67.84	38.18	33.63	26.73	23.83	0	13.16	29.24	21.77	15.50	11.87	13.84
Employment																
EMPL	3,050	3,665	3,570	3,550	3,790	3,899	3,890	2,900	2,590	2,375	1,655	1,522	1,519	1,317	1,320	1,310
Dividend																
DIVSAL (%)	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	7.36	8.31	11.34	14.95	16.77	19.762010 27	21.98516 093
DPS (TZS)	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	20.37	37.72	58.36	73.75	85.27	90.729629 05	107.0005 239
Earnings																

Table 4.1: TBL's Pre- and Post-Privatisation Financial and Operating Performance Metrics

Asian Institute of Research

FPS (TZS) -12 36 -13 65 -14 58	-20.97 -0.25 -16.43	1.81 -10.92 -5.79 3	5 65 61 22 61 64 72 05	130.87574
110 (110) 12:30 13:03 11:30	20.77 0.25 10.15	101 10.52 5.77 5	105 01122 01101 12105	20 117.77

Source: TBL's Annual Reports and calculations therefrom

Table 4.2: TBL's Descriptive Statistics

	Performanc e Metrics	Mini	Minimum		Maximum		Mean		Variance		Pearson Correlatio n	t Stat	P(T<=t) one-tail	P(T<=t) two-tail	t Critica l one- tail	t Critic al two- tail
	e Metrics	Pre	Post	Pre	Post	Pre	Post		Pre	Post						
1	Profitability															
(i)	ROS (in %)	-223.06	-4.31	5.85	28.51	-118.80	15.05	98.23821	10345. 6	96.821175 2	0.7323097 24	3.9958 48	0.0026087 12	0.0052174 24	1.8945 7861	2.3646 243
(ii)	ROA (in %)	-23.55	-3.99	1.41	31.33	-12.74	21.94	20.51820	90.191 8	124.61300 9	0.4791905 6	9.2197 44	0.0000182 27	0.0000364 53	1.8945 7861	2.3646 243
(iii)	ROE (in %)	-43.90	-5.54	2.24	47.79	-24.13	35.17	35.31687	357.35 5	305.77891 0	0.5206379 47	9.3923 46	0.0000161 46	0.0000322 93	1.8945 7861	2.3646 243
2	Efficiency															
(i)	SALEFF (in TZS M)	0.33697135	11.1903474	1.91206 897	91.11617 312	0.977398 41	63.268760 52	38.03666	0.3650 767	882.62970	0.7721713 46	6.0244 93	0.0002645 96	0.0005291 93	1.8945 7861	2.3646 243
(ii)	NIEFF (in TZS M)	1.27605634	-0.4826254	0.10025 7069	23.39318 182	0.684582 44	11.582969 11	8.597570	0.2246	72.175607	0.4324407	4.1796 36	0.0020695 64	0.0041391	1.8945 7861	2.3646 243
3	Capital Expenditure															
(i)	CESA (in %)	47.42	3.69	166.67	27.71	98.59	100.61	55.15433	2222.4 252	68.339012 4	0.7477517 77	5.8868 78	0.0003037 39	0.0006074 78	1.8945 7861	2.3646 243
(ii)	CETA (in %)	8.84	0.80	15.89	26.38	12.54	14.25	6.338218	3.9738 889	80.447267 5	0.1635688 56	0.5444 91	0.3015040 613	0.6030081 225	1.8945 7861	2.3646 243
4	Output														_	_
(i)	PROD (in Hectolitres)	435,347	575,000	531,127	1,551,00 0	495,734. 38	1,281,626	478,308	1.48E+ 09	135,832,0 00,000	0.6232275 27	- 5.6462 97	0.0003887 352	0.0007774 70	1.8945 7861	2.3646 243
(ii)	RESA (in TZS M)	39.79	6,495.09	739.70	62,434.1 6	329.21	41,470.08	25,464.27	81,980. 13	422,225,2 51.77	0.8321954 86	5.7292 60	0.0003567 39	0.0007134 8	1.8945 7861	2.3646 243
5	Financial Leverage														_	_
(i)	TDTA (%)	0.00	0.00	79.24	17.14	28.62	9.98	20.66337	689.35 3	27.104167 09	0.8214009	2.3767 97	0.0245564 44	0.0491128 88	1.8945 7861	2.3646 243
(ii)	TDTE (%)	0.00	0.00	159.17	29.24	51.29	16.15	40.72436	2768.8 0	79.409970 13	0.7563470 36	2.1491 44	0.0343501 94	0.0687003 88	1.8945 7861	2.3646 243
6	Employment															
(i)	EMPL	2,900	1,310	3,899	2,590	3,539	1,701	1,041	140,02 9	251,054	0.3207657 15	7.2708 17	8.34514E- 05	0.0001669 03	1.8945 7861	2.3646 243
7	Dividend Payout														_	_
(i)	DIVSAL (in %)	0.00	0.00	0.00	21.99	-	12.56	8.16	0.00	52.63	No correl	4.8963 86	0.0008801 83	0.0017603 67	1.8945 7861	2.3646 243
(ii)	DPS (in TZS)	0.00	0.00	0.00	107.00	- .00	59.15	39.70	0.00	1,377.87	No correl	4.5071 54	0.0013872 28	0.0027744 56	1.8945 7861	2.3646 243
8	Earnings														_	_
(i)	EPS (in TZS)	-20.97	-5.79	1.81	130.88	-10.92	71.29	52.50	61.44	1982.29	0.40	-5.54	0.0004360 084	0.0008720 17	1.8945 7861	2.3646 243

Source: Author's calculations

Table 4.2 indicates that the pre-privatisation minimum ROA was -23.55% while the post-privatisation minimum ROA was -3.99%. Table 4.2 also indicates that the pre-privatisation maximum ROA was 1.41% whereas the post-privatisation maximum ROA was 31.33%. Table 4.2 shows that the average pre-privatisation ROA stood at -12.74% and the average post-privatisation ROA was 21.94%. The *t*-stat of -9.219744, t-critical value (one-tail) of 1.89457861 and the p-value (one-tail) of 0.000018227 (which is less than significance level of 0.05) (refer to table 4.2) provide strong evidence to reject the null hypothesis. This means that the difference between the average ROA prior to privatisation and the average ROA after privatisation (increase) is statistically significant. In other words, privatisation of TBL had significant positive effect on return on assets.

Table 4.2 indicates that the pre-privatisation minimum ROE was -43.90% and the post-privatisation minimum ROE was -5.54%. Table 4.2 also indicates that the pre-privatisation maximum ROE was 2.24% whereas the

post-privatisation maximum ROE was 47.79%. Table 4.2 shows that the average pre-privatisation ROE stood at -24.13% while the average post-privatisation ROE was 35.17%. Table 4.2 also shows the *t*-statistic of -9.392346, t-critical value (one-tail) of 1.89457861 and the p-value (one-tail) of 0.000016146 (which is less than the significance level of 0.05). These results provide strong evidence that TBL privatisation resulted into significant improvement (increase) in ROE. Therefore, null hypothesis is rejected on this basis because there is strong evidence to do so.

4.3. Change in Operating Efficiency after Privatisation

The minimum pre-privatisation sales per employee (SALEFF) was TZS 336,971.35 whereas the minimum postprivatisation SALEFF was TZS 11,190,347.49 (refer to Table 4.2). Table 4.2 also indicates that the preprivatisation maximum SALEFF was TZS 1,912,068.97 whereas the post-privatisation maximum SALEFF was TZS 91,116,173.12. The pre-privatisation average SALEFF was TZS 977,398.41 whereas the post-privatisation average SALEFF was TZS 63,268,760.52 (refer to Table 4.2). With *t*-statistic of -6.024493, t-critical value (onetail) of 1.89457861 and p-value (one-tail) of 0.000264596 (which is less than significance level of 0.05), there is strong evidence that difference between average SALEFF after privatisation is higher than that before privatisation. The null hypothesis is rejected. TBL privatisation resulted into significant improvement of sales efficiency.

Table 4.2 indicates that the pre-privatisation minimum net income per employee (NIEFF) was TZS - 1,276,056.34 and the post-privatisation minimum NIEFF was TZS -482,625.48. The pre-privatisation maximum NIEFF was TZS 100,257.07 and post-privatisation maximum NIEFF was TZS 23,393,181.82 (Table 4.2). The average pre-privatisation NIEFF was TZS -684,582.44 while the average post-privatisation NIEFF was TZS 11,582,969.11 (Table 4.2). Table 4.2 also shows that *t*-statistic of -4.179636, t-critical value (one-tail) of 1.89457861 and p-value (one-tail) was 0.002069564 (which is less than significance level of 0.05). These statistics provide strong evidence to reject the null hypothesis. This means that the difference between average NIEFF before privatisation and average NIEFF after privatisation is statistically significant. Privatisation of TBL had significant positive effect on net income efficiency.

4.4. Change in Capital Expenditure after Privatisation

The minimum pre-privatisation capital expenditure to sales (CESA) was 47.42% whereas the minimum postprivatisation CESA was 3.69% (refer to Table 4.2). Table 4.2 also shows that the pre-privatisation maximum CESA was 166.67% while the post-privatisation maximum CESA was 27.71%. The pre-privatisation average CESA was 98.59% whereas the post-privatisation average CESA was 100.61% (refer to Table 4.2). The *t*statistic of 5.8868780, t-critical value (one-tail) of 1.89457861 and the p-value (one-tail) of 0.000303739 (which is less than significance level of 0.05) provide strong evidence that the difference between average postprivatisation CESA is higher than the average pre-privatisation CESA. Thus, null hypothesis is rejected. TBL privatisation resulted into significant improvement (increase) of capital expenditure.

The minimum pre-privatisation capital expenditure to assets (CETA) was 8.84% whereas the minimum postprivatisation CETA was 0.80% (refer to Table 4.2). Table 4.2 also shows that the pre-privatisation maximum CETA was 15.89% whereas the post-privatisation maximum CETA was 26.38%. The pre-privatisation average CETA was 12.54% whereas the post-privatisation average CETA was 14.25% (refer to Table 4.2). The *t*-statistic of -0.544491, t-critical value (on-tail) of 1.89457861 and the p-value (one-tail) of 0.3015040613 is higher than the significance level of 0.05. These statistics provide weak evidence that an increase in average CETA resulted from TBL's privatisation. Null hypothesis is not rejected on this basis.

4.5. Change in Output after Privatisation

Table 4.2 indicates that the pre-privatisation minimum units produced (in hectolitres) (PROD) was 435,347 hectolitres whereas the post-privatisation minimum PROD was 575,000 hectolitres. The pre-privatisation maximum PROD was 531,127 hectolitres whereas post-privatisation maximum PROD was 1,551,000 hectolitres (refer to Table 4.2). The average pre-privatisation PROD was 495,734.38 hectolitres while the average post-

privatisation PROD was 1,281,626 hectolitres (refer to Table 4.2). Table 4.2 shows that *t*-statistic was - 5.646297, t-critical value (one tail) was 1.89457861 and p-value (one-tail) was 0.0003887352 (which is less than the significance level of 0.05). These statistics provide strong evidence that the difference between average PROD before privatisation and average PROD after privatisation (increase) is statistically significant Thus, null hypothesis is rejected because there is strong evidence to do so. In other words, privatisation of TBL had significant improvement on production.

The minimum pre-privatisation real sales revenue (RESA) was TZS 39,788,280 whereas the minimum postprivatisation RESA was TZS 6,495,090,300 (refer to Table 4.2). Table 4.2 also shows that pre-privatisation maximum RESA was TZS 739,703,000 while post-privatisation maximum RESA was TZS 62,434,158,400. The pre-privatisation average RESA was TZS 329,213,905 whereas the post-privatisation average RESA was TZS 41,470,078,950 (Table 4.2). The *t*-statistic of -5.729260, t-critical value (one tail) of 1.89457861 and p-value (one-tail) of 0.000356739 (which is less than significance level of 0.05) provide strong evidence that sales in real terms improved (increased) significantly after privatisation. Thus, null hypothesis is rejected.

4.6. Change in Financial Leverage after Privatisation

Table 4.2 indicates that the pre-privatisation minimum debt to assets (TDTA) ratio was 0.00% and the postprivatisation minimum TDTA was 0.00%. The pre-privatisation maximum TDTA was 79.24% and the postprivatisation maximum TDTA was 17.14% (see table 4.2). The average pre-privatisation TDTA was 28.62% and the average post-privatisation TDTA was 9.98% (see table 4.2). Table 4.2 shows that the *t*-statistic was 2.376797, t-critical value (one tail) was 1.89457861 and p-value (one-tail) was 0.024556444 (i.e., less than the significance level of 0.05). The statistics show strong evidence that privatisation resulted to significant improvement (decline) in debt to asset ratio. Therefore, null hypothesis is rejected on this basis.

Table 4.2 indicates that the pre-privatisation minimum debt to equity (TDTE) ratio was 0.00% and the postprivatisation minimum TDTE was 0.00%. The pre-privatisation maximum TDTE was 159.17% and the postprivatisation maximum TDTE was 29.24% (see table 4.2). The average pre-privatisation TDTE was 51.29% and the average post-privatisation TDTE was 16.15% (see table 4.2). With *t*-statistic of 2.149144, t-critical value (one tail) of 1.89457861 and p-value (one-tail) of 0.034350194 (which is less than the significance level of 0.05), there is strong evidence to show that privatisation of TBL resulted into statistically significant improvement (decline) in debt-to-equity ratio. Thus, null hypothesis is rejected.

4.7. Change in Level of Employment after Privatisation

The minimum pre-privatisation employment level (in terms of number of employees) (EMPL) was 2,900 employees whereas the minimum post-privatisation EMPL was 1,310 employees (refer to table 4.2). Table 4.2 also shows that the pre-privatisation maximum EMPL was 3,899 employees and the post-privatisation maximum EMPL was 2,590 employees. The pre-privatisation average EMPL was 3,539 employees and the post-privatisation average EMPL was 1,701 employees (refer to table 4.2). With *t*-statistic of 7.270817, t-critical value (one tail) of 1.89457861 and p-value (one-tail) of 0.0000834514354 (i.e., less than the significance level of 0.05), there is strong evidence that employment level declined significantly after TBL's privatisation move. The null hypothesis that the change in EMPL is zero is rejected on this basis.

4.8. Change in Dividend Payout after Privatisation

Table 4.2 indicates that the pre-privatisation minimum dividend to sales (DIVSAL) ratio was 0.00% and the post-privatisation minimum DIVSAL was 0.00%. The pre-privatisation maximum DIVSAL was 0.00% and the post-privatisation maximum DIVSAL was 21.99% (refer table 4.2). The average pre-privatisation DIVSAL was 0.00% and the average post-privatisation DIVSAL was 12.56% (see table 4.2). Table 4.2 also shows that *t*-

statistic was -4.896386, t-critical value (one tail) was 1.89457861 and p-value (one-tail) was 0.000880183 (i.e., less than the significance level of 0.05). The data show that privatisation of TBL resulted into statistically significant increase in dividend to sales ratio. Therefore, the null hypothesis is rejected.

Table 4.2 indicates that the pre-privatisation minimum dividend per share (DPS) was TZS 0.00 and the postprivatisation minimum DPS was TZS 0.00. The pre-privatisation maximum DPS was TZS 0.00 and the postprivatisation maximum DPS was TZS 107.00 (refer table 4.2). The average pre-privatisation DPS was TZS 0.00 and the average post-privatisation DPS was 59.15 (see table 4.2). With *t*-statistic of -4.507154, t-critical value (one tail) of 1.89457861 and p-value (one-tail) of 0.001387228 (which is less than the significance level of 0.05), there is strong evidence to indicate that privatisation of TBL resulted into statistically significant improvement (increase) in dividend per share paid to shareholders. Thus, null hypothesis is rejected.

4.9. Change in Earnings after Privatisation

The minimum pre-privatisation earnings per share (EPS) was TZS -20.97 and the minimum post-privatisation EPS was TZS -5.79 (see to table 4.2). Table 4.2 also shows that the pre-privatisation maximum EPS was TZS 1.81 and the post-privatisation maximum EPS was TZS 130.88. The pre-privatisation average EPS was -10.92 and the post-privatisation average EPS was TZS 71.29 (see table 4.2). Table 4.2 also shows that *t*-statistic was - 5.54, t-critical value (one tail) was 1.89457861 and p-value (one-tail) was 0.0004360084 (which is less than the significance level of 0.05). These statistics show that there is strong evidence that EPS improved (increased) significantly after TBL's privatisation. Therefore, null hypothesis that a change in EPS is zero is rejected.

Tables 4.1 and 4.2 generally indicate that there was post-privatisation improvement for all performance metrics except capital expenditure to asset ratio (CETA).

5. Concluding Remarks and Areas for further Researches

Extant theoretical and empirical evidence generally supports the common-place belief that SOEs are less efficient and less profitable than similar privately-owned firms. The author carried out the t-test Paired Two Sample for Means data analysis (using Microsoft Excel Data Analysis Tool). The results from the TBL case study showed that financial and operating performance metrics improved after privatisation. The financial and operating performance metrics improved after privatisation. The financial and operating performance metrics that showed statistically significant improvement are return on sales (ROS), return on assets (ROA), return on equity (ROE), sales efficiency (SALEFF), net income efficiency (NIEFF), capital expenditure to sales (CESA), units produced (in hectolitres) (PROD), real sales (RESA), debt to assets (TDTA), debts to equity (TDTE), employment levels (in terms of number of employees) (EMPL), dividend to sales (DIVSAL), dividend per share (DPS) and earnings per share (EPS). However, results did not provide sufficient evidence to show that privatisation improved capital expenditure (in terms of capital expenditure to assets) (CETA) ratio. Thus, CETA might have improved by chance, not as a result of privatisation.

The current study did not explore multi-factor model(s) which affect corporate performance of a privatised firm. It is suggested that further studies be carried out to explore the effect of a simultaneous combination of multi-factors on corporate performance. The multi-factor model(s) to be studied may include macro-economic conditions like economic cycle, that is, whether economy is in recession, expansion or boom), exchange rates trends, interest rates trends, availability of foreign exchange and the like and their effect on post-privatisation financial and operating performance. The current study also did not make an attempt to study the effect of privatisation to consumers. In view of these gaps, it is suggested that further studies be carried out on the effect of privatisation to consumers in Tanzania.

The study also did not embark on the effect of reforms (other than privatisation) on the financial and operating performance of SOEs in Tanzania. Those non-privatisation reforms include reduction on barriers to trade, fiscal measures, etc. In view of this, it is suggested that further studies be carried out on "substitutes" for privatisation for financial and operating performance improvement in Tanzania.

Funding: Not applicable.

Conflict of Interest: The authors declare no conflict of interest.

Informed Consent Statement/Ethics Approval: Not applicable. References

- Al Hinai, Mansour Talib (2016), The Impact of Privatization on Company Performance: The Case of Egypt, PhD Thesis, University of Birmingham.
- Arcas, Maria-Jose and Patricia Bachiller (2010), "Operating Performance of Privatized Firms in Europe: Organizational and Environmental Factors", *International Journal of Public Administration*, Vol. 33, No. 10, pp 487-498.
- Arocena, Pablo and Oliveros, Diana, (2012), "The efficiency of state-owned and privatized firms: Does ownership make a difference?", *Elsevier*, Vol. 140, Issue No. 1, pp 457-465.
- Bachiller, Patricia (2012), "The Impact of Privatization on Economic Performance in European Companies", Journal des Économistes et Études Humaines, Vol. 18, No. 1.
- Bachiller, Patricia (2017), "A Meta-Analysis of the Impact of Privatization on Firm Performance", *Management Decision*, Vol. 55, No. 1, pp 178-202.
- Balza, Lenin H., Raul Alberto Jimenez Mori, Jorge Enrique Mercado Díaz (2013), "Privatisation, institutional reform, and performance in the Latin American electricity sector", Technical Note No. IDB-TN-599, Inter-American Development Bank.
- Bartolotti, B., J. D'Souza, M. Fantini and W.L. Megginson (2001) "Sources of Performance Improvement in Privatised Firms: A Clinical Study of the Global Telecommunications Industry", *Social Science Research Network Electronic Paper Collection*.
- Buckland, R. and P. Fraser (2002), "The Scale and Patterns of Abnormal Returns to Equity Investment in UK Electricity Distribution", *The Global Finance Journal*, Vol. 13, No. 1. pp 39-62.
- Chen, Yuyu; Mitsuru, Igami; Masayuki, Sawada and Mo Xiao (2021), "Privatization and productivity in China", *The RAND Journal of Economics*, Vol. 52, No. 4, pp 884-916.
- Čučković, Nevenka; Krešimir, Jurlin and Valentina Vučković (2011), "The Privatisation of Public Utilities in Croatia: As Assessment of the Major Gains and Pains", *LSEE Papers on South Eastern Europe*, Issue 3.
- D'Souza J., W. L. Megginson and R. Nash (2005), "Effects of Institutional and Firm-specific Characteristics on post-privatization Performance: Evidence from Developed Countries, *Journal of Corporate Finance*, Vol. 11, No. 5, pp. 747-766.
- Dewenter, K. and P. H., Malatesta (2001), "State-owned and Private-owned Firms: An empirical Analysis of Profitability, Leverage and Labour Intensity, *American Economic Review*, Vol. 91. No. 1, pp. 320-334.
- Earle, John S. and Solomiya Shpak (2019), "Impact of privatization on employment and earnings", *IZA World of Labor*, 93v2.
- Estrin, Saul and Adeline Pelletier (2018), "Privatization in Developing Countries: What Are the Lessons of Recent Experience?", *The World Bank Research Observer*, Vol. 33, No. 1, pp 65-102.
- Kaare, Summa (2002), "The Role of Privatisation in Providing the Urban Poor Access to Social Services: The Case of Solid Waste Collection in Dar es Salaam, Tanzania", Research on Poverty Alleviation (REPOA), Research Report No. 02.2.
- Kabelwa, G. (2002), "South African FDI into East Africa: The Case of Tanzania", Economic and Social Research Foundation (ESRF), a paper prepared for the Globalization and East Africa Project and delivered as Paper 3 to the Project's Seminar in Oct. 30, 2002.
- Kenton, Will (2022), "Base-Year Analysis: What it Means, How it Works.
- Kisenge, Frank Fredrick (2013), The Impact of privatizing Banking Industry in Tanzania: A Case Study of National Bank of Commerce. A Master Dissertation Mzumbe University.
- Kouser, Rehana; Toseef Azid and Karamat Ali (2011), "Reasons for Privatization and Consequent Role of Government: Comprehensive Study Based on Early Evidence", *International Journal of Contemporary Business Studies*, Vol. 2, No. 10, pp 35-52.
- Megginson W. L., and J. M. Netter (2001), "From State to Market: A Survey of Empirical Studies on Privatization", *Journal of Economic Literature*, Vol. 39, No. 2, pp. 321-389.
- Mongula, Benedict (2002), "Performance of Privatised Firms in Tanzania", *Business Management Review*, Vol. 8, Issue No. 2, pp 81-107.
- Muganda, A. (2004), "Tanzania's Economic Reforms-and Lessons Learned: A Case Study from Reducing Poverty, Sustaining Growth-What Works, What Doesn't, and Why a Global Exchange for Scaling Up Success: Scaling Up Poverty Reduction. A Global Learning Process and Conference Shanghai May 22-25, 2004.

- Portelli, B. and R. Narula (2004), "Foreign Direct Investment through Acquisitions and Implications for technological Upgrading: Case Evidence from Tanzania", *MERIT-Infonomics Research Memorandum Series*.
- Radić, Mislav; Davide Ravasi and Kamal Munir (2021), "Privatization: Implications of a Shift from State to Private Ownership", *Journal of Management*, Vol. 47, No. 6, pp 1596-1629
- Redda, Bereket Mehari (2007), Post-privatisation Changes in Management Control, Firm Activities and Performance: The Case of Eritrea-based firms, PhD. Thesis, University of Groningen
- Sancheti, D.C and V.K. Kapoor (2007), Statistics [Theory, Methods & Application], 7th Thoroughly Revised Edition, Sultan Chand & Sons, New Delhi
- Shirley, Mary M. (1992), "The What, Why, and How of Privatization: A World Bank Perspective", *Fordham Law Review*, Vol. 60, No. 6, pp 23-36.
- Waigama, Samwel M. S. (2008), "Privatization Process and Asset Valuation: A Case Study of Tanzania", Doctoral Thesis in Building and Real Estate Economics, School of Architecture and the Built Environment, Royal Institute of Technology.
- World Bank (2023), World Development Indicators: Employment, Governance and Policy, Governmental Statistics, Infrastructure, Public Health, Social for Tanzania